Author's response to reviews

Title: Protocol for a systematic review and meta-analysis of cognitive-behavioural therapy for social anxiety disorder in psychosis

Authors:

Maria Michail (maria.michail@nottingham.ac.uk)
Max J Birchwood (M.J.Birchwood@warwick.ac.uk)
Lynda Tait (Lynda.Tait@nottingham.ac.uk)

Version: 3 Date: 15 May 2014

Author's response to reviews:

Dear Editor,

Please find the revised version of the manuscript "Protocol for a systematic review and meta-analysis of cognitive-behavioural therapy for social anxiety disorder in psychosis".

We have made the following changes as requested:

1. We have amended the title of the manuscript to reflect that this is a protocol
2. We have specified that we will be reporting the outcomes of the review according to PRISMA
3. We have cited the methodological guidance we will be following (Cochrane Handbook)

Yours sincerely,

Dr Maria Michail